Author's response to reviews

Title: Health Insurance, Neighborhood Income, and Emergency Department Usage by Utah Children 1996-1998

Authors:

Anthony Suruda (tonysuruda@aol.com)
Thomas J Burns (tburns@ou.edu)
Stacey Knight (Stacey.Knight@hsc.utah.edu)
J. MICHAEL Dean (Mike.Dean@hsc.utah.edu)

Version: 5 Date: 10 March 2005

Author's response to reviews:

I submitted the revised manuscript with the requested formatting changes yesterday. There was an error in the second paragraph of the discussion. I have changed the text to read "approximately 50% greater than" instead of "approximately twice" and this is the only change made.

Anthony Suruda